Author’s response to reviews

Title: Alcohol use among university students in Sweden measured by an electronic screening instrument

Authors:

Agneta Andersson (agneta.andersson@liu.se)
Ann-Britt Wiréhn (ann-britt.wirehn@liio.se)
Christina Ölvander (christina.olvander@logica.com)
Diana Stark Ekman (dianastarkekman@gmail.com)
Preben Bendtsen (preben.bendtsen@liu.se)

Version: 8 Date: 6 July 2009

Author’s response to reviews:

We have made some minor changes in the layout of the tables and sub-headings.